Author's response to reviews

Title: Ultrasound settings significantly alter arterial lumen and wall thickness measurements

Authors:

Kathleen Potter (pottek04@cyllene.uwa.edu.au)
Christopher J Reed (Christopher.Reed@health.wa.gov.au)
Daniel J Green (d.j.green@ljmu.ac.uk)
Graeme J Hankey (gihankey@cyllene.uwa.edu.au)
Leonard F Arnolda (Leonard.Arnolda@health.wa.gov.au)

Version: 2 Date: 11 January 2008

Author's response to reviews:

Hi,

The paper has been revised according to instructions:

1. Title page has been left-justified.
2. References have been placed before the full-stop.
3. Competing interests section reworded according to instructions.
4. Author's contribution section reworded according to instructions.
5. Figures renamed as "Figure legends"
6. Additional files cited within manuscript text according to instructions

regards,
Kathleen Potter